

Federal Reserve Bank of Chicago

Bank Imputed Interest Rates: Unbiased Estimates of Offered Rates?

Evren Ors and Tara Rice

WP 2006-26

Bank Imputed Interest Rates: Unbiased Estimates of Offered Rates?

Evren Ors Finance & Economics Department HEC School of Management 78351 Jouy-en-Josas, France +33.1.39.67.71.23 ors@hec.fr

and

Tara Rice Economic Research Federal Reserve Bank of Chicago 230 South LaSalle Street Chicago, Illinois, USA 312-322-5274 trice@frbchi.org

November 2006

Abstract

We examine whether "imputed" interest rates obtained from bank financial statements are unbiased estimates of "offered" interest rates that the same banks report in surveys. We find evidence of a statistically significant amount of bias. However, the statistical bias that we document does not appear to be economically significant. When used as dependent variables in regression analysis, imputed rates and offered rates lead to the same policy conclusions. Our work has important methodological implications for empirical research that examines the product market competition among depository institutions.

Key words: deposit rates, transactional rates, imputed prices, product market, competition

JEL codes: G21, L11

The views presented here are those of the authors and do not necessarily reflect the views of the Federal Reserve Bank of Chicago or the Federal Reserve System. We are thankful for comments and suggestions made by Robert DeYoung, Bill Marcum, session participants at the 2006 WEA Meetings, and the seminar participants at the Federal Reserve Bank of Chicago, the Federal Deposit Insurance Corporation, George Washington University, and the Board of Governors of the Federal Reserve System.

Bank Imputed Interest Rates: Unbiased Estimates of Offered Rates?

1. Introduction

Analysis of product market competition in the banking and finance literature (and more generally, in the industrial organization literature) has critical public policy implications. It aids antitrust analysis, for example, in determining whether proposed bank mergers will adversely affect consumers through increased market concentration. The theory models on which this analysis is based refer to prices that the banking firm offers to its customers given the structure of the market it faces. Likewise, the empirical literature examines the statistical relationship between the banking product prices and market concentration. This line of research uses deposit and loan interest rates as dependent or explanatory variables. Those prices are, however, a topic of contention. Prices are difficult to measure and in the banking research, as is the case in most fields of economic research, transaction prices are difficult to obtain. The alternative is to impute prices, that is, to calculate prices by taking the ratio of interest expenses (revenue) to the stock of deposits (loans). However, imputed prices, which incorporate measurement error, may incorporate bias that affects the results of empirical analyses.

In this paper, we examine whether "imputed" (calculated) interest rates are suitable estimates of "offered" (transaction) interest rates using a set of U.S. banks for which both rates are observable. What we term as "offered" rates are those explicitly stated by banks in surveys, and we assume that the offered rates are indeed the "true" interest rates to which the theory models refer.¹ Many researchers, however, rely on interest rates that are imputed from the financial statements because offered rates are either only observable for a limited sample of banks or simply not observable. Even when observed, offered rates may not be readily accessible because they are collected through surveys that are either confidential or sold by private data providers.² This contrasts with imputed interest rates which are calculated using

¹ We discuss whether such an assumption is warranted later in the paper.

² In the U.S., interest rate survey data collected by the Federal Reserve System, such as the Monthly Survey of Selected Deposits or the Survey of the Terms of Lending, remain confidential. Access is limited to economists working for one of the federal banking regulators. Data from private data providers, such as the Bank Rate Monitor, are available at a non-trivial cost.

information provided in the financial statements. While financial statements, and as a result, imputed interest rates, are available at no cost for the population of U.S. banks, these constructed variables may involve significant measurement error that could lead to systematic bias. Whether such bias exists and, if it exists, whether it affects empirical model estimates are questions that have not been systematically explored.³ In this paper, our goal is to determine whether using imputed rather than offered interest rates leads to any statistically and economically significant bias. We focus on deposit accounts, rather than loans, because the former have more standard product characteristics than the latter.⁴

Our work has important methodological implications for large strands of empirical research on depository financial institutions. Our results provide evidence of statistically significant bias: the raw imputed deposit rates are not particularly good estimates of the stated interest rates that banks offer to their customers. However, the observed bias decreases when imputed rates are winsorized or truncated, with the truncation being more effective than the winsorization. Once the imputed rate data have been truncated, they become more suitable estimates of the offered rates. Moreover, the observed statistical bias is not economically significant. We show that the coefficient estimates obtained when either imputed or offered rates are used as dependent variables have the same sign, same order of magnitude and the same order of statistical significance. The important implication of our work is that, when properly adjusted for measurement error, imputed interest rates lead to the same policy conclusions as offered rates. The paper proceeds as follows: section 2 presents the data, section 3 provides a short survey of the relevant literature, section 4 presents our empirical analysis, and section 5 concludes.

³ Hannan and Prager (1999) touch on this in their work. We discuss their paper below.

⁴ Loans are significantly more heterogeneous intermediation products than deposits. This holds true even for more standardized lending contracts, such as mortgages. Although it is possible to observe and control for some of the price and non-price loan characteristics in the loan survey data that are available (such as the Survey of the Terms of Lending), it is not possible to create imputed rates for homogeneous loan categories using Call Reports due to lack of detail at the product level. Moreover, loans' differing maturities make it difficult to compare imputed loan rates with lending rates offered to customers.

2. Data on Offered Rates and Imputed Rates

We combine data from three different sources. Imputed deposit interest rates are calculated using data from the Consolidated Reports of Condition and Income (the Call Reports), the publicly available quarterly financial statements that all U.S. commercial banks file with their federal regulators. We impute rates for three types of deposit accounts for which information is available in the Call Reports: Negotiable Order of Withdrawal (NOW) accounts, Money Market Deposit Accounts (MMDA), and passbook savings accounts. NOW accounts, first introduced in 1980, are interest paying withdrawal-upon-demand checking accounts with minimum balance requirements. MMDA accounts, first introduced in 1982 by banks to compete with the Money Market Mutual Funds, pay higher interest rates than NOW accounts but have more restrictive check-writing privileges and higher minimum balance requirements than NOW accounts.⁵ Passbook savings accounts are traditional household saving instruments without check-writing privileges.⁶

The imputed rates are calculated for each of the three account types by dividing the quarterly interest expense in the Call Reports by the stock of deposits for that quarter. It should be noted that the Call Report deposit expense items do not include any other fees or expenses related to deposit account management.⁷ We use two measures of quarterly imputed rates. The first of these is based on the latest stock of deposits of the corresponding account type at the current quarter's end, and the second is based on the average of deposits over the quarter.⁸ The survey offered rates are stated as simple (non-

⁵ Prior to the introduction of NOW and MMDA accounts, checking accounts did not pay explicit interest in the U.S. By 1984, the start of our sample period, both types of accounts were among the standard banking products.

⁶ We exclude certificates of deposits (CDs) and negotiable certificates of deposits (NCDs), which are non-checkable savings accounts with specific maturities, from our analysis. Although interest rates on these instruments are included in the MSSD survey, there are no corresponding interest expense items in the Call Reports that are detailed enough to impute maturity-specific CD rates.

⁷ Our focus, similar to most of the empirical banking literature, is on explicit interest rates. Implicit rates paid on deposits (in the form of service flows, such as the number checks or transfers allowed per month) and fees that depositors are charged are excluded from our analysis. Note that, since we match imputed rates that we calculate for a particular bank with the offered rates by the same institution, differences in implicit interests or fees charged across banks do not affect our analysis.

⁸ The stock of deposits is stated in Schedule RC-E of the Call Reports and the quarterly average is reported in Schedule RC-K of the Call Reports.

compounded) annual percentage rates. To make the quarterly imputed rates comparable with the annual offered rates, we multiply the quarterly imputed rate by four to obtain a simple annual rate.⁹

NOW, MMDA and savings account interest rates offered by banks are collected from the Federal Reserve System's Monthly Survey of Selected Deposits (MSSD), available to us starting with January 1987. The MSSD stopped collecting offered interest rate information in September 1994 and was discontinued in 1997.¹⁰ The survey respondents were asked to provide the most commonly offered interest rate on the largest volume of deposits per deposit type during the last week of each month.¹¹ The MSSD provided offered-rate data at the bank rather than at the bank and market (or branch) level. Because the MSSD data sampling frequency is higher than the reporting frequency of the Call Reports, we use two alternative measures: the survey rates from the last month of the quarter and the average of monthly survey rates for the quarter.

We also collect market structure variables from the Federal Deposit Insurance Corporation's (FDIC) annual Summary of Deposits datasets, and market level personal income data from the Bureau of Economic Analysis' Regional Economic Information database.

An additional survey data set is currently available to researchers, the Bank Rate Monitor (BRM) data. The BRM is a survey provided by a private data vendor that collects bank and market level data on a weekly basis. Due, however, to substantial reporting issues which render these deposit prices incomparable with Call Report imputed prices, we do not use the BRM data. The primary issue is that the BRM Survey asks participating banks to provide the rates for the banks' "lowest minimum to open" non-interest checking account and "lowest minimum to open and earn interest" checking account. Thus, the

⁹ Note that the resulting imputed rate is based on interest paid, which is compounded interest. Since the Call Reports do not include any information regarding the frequency of the compounding periods (which varies across banks from monthly to semi-annual) we do not make any further adjustments to imputed rates. As it will be seen below, despite this error, our imputed rates slightly underestimate the offered survey rates.

¹⁰ Until end of June 1989, MSSD collected the most commonly offered rate per account type. Starting with July 1989, MSSD surveys collected more detailed NOW and MMDA information that accounted for the possibility that the bank may offer higher rates for larger maintained-balances. For consistency in our data series, we filtered these data to collect only the most commonly observed offered rate per account type. Samples available to us end in September 1994, because after that date MSSD started to collect interest *expense* information, instead of interest *rate* information.

¹¹ Arguably, offered interest rate per *number* of accounts is a better definition for "most common" than one based on the dollar volume of deposits, but MSSD opts for the latter rather than the former.

rates stated in the BRM data are effectively the lowest rates offered by banks, not the most-commonly cited rates (the mode) as in the MSSD data or the imputed average interest rate paid by the bank (the mean) obtained from the Call Report data. Moreover, BRM does not provide information on the percentage of deposits that earn the lowest rates, nor information on rates paid to accounts with greater deposit balances. As a result, we cannot infer the mean or the mode of the rate distribution from BRM data. In summary, as no additional information on the distribution of the BRM deposit rates is available that would allow researchers to infer the central tendencies of the surveyed rates, we conclude that the use of BRM data in analysis would provide ambiguous results.¹²

3. Literature Survey

Given the methodological focus of this paper, we confine our literature survey to the studies, namely the Structure Conduct Performance (SCP) tests, that have examined the relationship between local market concentration and deposit interest rates where either the offered or the imputed interest rates have been used as dependent variables. In section 4.3, we test whether using imputed versus offered rates in tests of market concentration and deposit rates leads to economically different results. The SCP paradigm implies a relationship between market concentration and firm conduct (in terms of profit or performance); such that noncompetitive behavior in more concentrated markets results in a positive (negative) relationship between market concentration and profitability (deposit prices).

While early studies examined the relationship between market concentration (as measured by the Hirschman Herfindahl Index) and profitability, Berger and Hannan (1989) was the first to test the pricemarket concentration relationship using MSSD offered rates rather than testing the profit-concentration relationship. The authors' results strongly support the SCP hypothesis when MMDA, NOW and short-

¹² We examined the BRM surveyed NOW and MMDA rates between January 1998 and December 2000 and compared them with the corresponding imputed rates obtained from the Call Reports. For the said period, BRM rates were significantly lower than the corresponding imputed rate. For NOW accounts mean (median) BRM rate was 1.23% (1.01%) compared to mean (median) Call Report imputed rate of 2.56% (1.64%) for the same banks over the same period. For MMDA accounts BRM rate mean was 2.29% (median 2.25%) compared to 3.14% (3.14%) for imputed rates.

term CD rates are used as measure of deposit account prices. Most of the studies that followed using the MSSD data confirm this negative relationship between deposit rates and market concentration (Calem and Carlino (1991), Berger and Hannan (1991)), though not without exceptions (for example, Jackson (1992)). Other studies using MSSD data examine how banks change the rates they offer depositors (Neumark and Sharpe (1992)), and whether the relationship between market concentration and deposit rates holds when concentration changes due to mergers (Prager and Hannan (1998)) or bank branching restrictions (Calem and Nakamura (1998)). A recent study by Brewer and Jackson (2004) considers the effects of including bank-specific risk variables (which capture nonperforming loans, capital and the interest-rate sensitivity of assets and liabilities) in the SCP analysis.

Another set of studies use imputed deposit rates obtained from the Call Report data (Heitfield and Prager (2002), Rosen (2003), Dick (2002), Hannan and Prager (2004) and Hannan and Prager (2006)) and find results generally consistent with studies using the MSSD data. These more recent studies contribute to the literature by testing additional aspects of the SCP paradigm. Rosen (2003), for example, examines whether the size distribution of banks in a market (in addition to market concentration) affects the deposit rates. Heitfield and Prager (1998) test the price-concentration relationship using alternative measures of markets (defined at the Metropolitan Statistical Area or MSA and state-level) and find that while local market concentration measures are useful, broader concentration measures are also appropriate. Several of the studies comment on the advantage of using the imputed Call Report prices given the extended time period for which these data are available (e.g., Rosen (2003), Heitfield and Prager (1998)).¹³

What differs among these studies is the particular interest rate that is found to have the strongest relationship (in terms of statistical significance), and the economic implications of the results. Berger and Hannan (1989) find that MMDA rates are 25 to 100 basis points less in the most concentrated markets than in the least concentrated markets. They find similar results for NOW and savings account rates, but

¹³ To date, only a few studies to date have used the Bank Rate Monitor data, partly due to its inaccessibility to researchers and academics until recently. These studies, however, have not been direct tests of the SCP hypothesis. Radecki (1998) and Heitfield (1999) examine the appropriate size of the 'local-market'; Rosen (2002) presents a model of price setting in the presence of heterogeneous customers; and Kiser (2003) studies whether conditions in the bank loan market affect the pricing of retail deposits.

not for longer-term CD rates. Calem and Carlino (1991) use MMDA and short-term CD rates (3 and 6 months) and find similar results, though they do not interpret the economic effects of their findings. Berger and Hannan (1991) focus solely on MMDA rates and find an asymmetric relationship between rate increases and decreases: a 29 basis point decrease in the rate of market treasuries leads to 62 percent probability that a bank will reduce its MMDA rate. The same increase in market rates leads to a 39 percent probability that a bank will increase its MMDA rate.

Neumark and Sharpe (1992) use MMDA and 6 month CD rates and find that the rates, on average, drop 60 basis points between the least to most concentrated markets. Rosen (2003) includes MMDA and NOW accounts also, and finds a small 4 to 7 basis point change in rates with a one standard deviation change in the market concentration. He attributes a larger change in deposit rates to market size than market concentration. Brewer and Jackson (2004) find that the magnitude of the relationship between deposit rates and market concentration decreases by 50 percent when bank-specific risk variables are included in the SCP analysis.

In summary, most studies use MMDA or NOW accounts, find strong statistical significance using MMDA accounts, less so using NOW accounts and find vastly different economic implications for the results. What is not evident is whether the different economic results found using different rates depend on the additional elements included in later studies (i.e., market size, megamergers) or the dataset and specific rates used. We examine this issue in tests of the SCP hypothesis below.

The only study that examines whether using MSSD offered rates versus Call Report imputed rates may lead to different inferences is Prager and Hannan (1999). In an earlier paper, Prager and Hannan (1998), the authors used MSSD *offered* rates and found that substantial horizontal mergers lead to statistically significant decreases in NOW and MMDA rates (24 and 32 basis point decreases, respectively). When revisiting this evidence using Call Report *imputed* rates (which results in significantly larger samples), Prager and Hannan (1999) find that large horizontal mergers have no effect on NOW and MMDA accounts, whereas such mergers lead to a small decrease in savings account rates (9

basis points).¹⁴ Prager and Hannan (1999) then investigate whether this discrepancy is due to the data source upon which each study relies. They repeat their analysis for a subsample of banks for which both offered and imputed rates are available, and find that coefficient estimates differ significantly depending on whether monthly MSSD offered rates or quarterly Call Report imputed rates are used. The authors note that the coefficient estimates have the same sign but they do not necessarily have the same statistical significance. Further, the authors find that the coefficients of correlation between offered and imputed series range between 0.731 and 0.933 for NOW accounts and 0.515 and 0.689 for MMDA accounts. The authors conclude Call Report imputed prices are "noisy" and "should be used with caution."

Our study differs from theirs in a number of dimensions. First, the Prager and Hannan (1999) analysis is limited to a comparison of the coefficient estimates of their model when offered or imputed rates are used as dependent variables. In contrast, we provide formal tests of the bias analyzing the direction and the size of the potential bias. Second, we discuss the reasons why the observed difference may exist in the two series. Third, Prager and Hannan (1999) use all of the monthly survey interest rate observations and the quarterly imputed series that are available to them when estimating their empirical model. In contrast, throughout our analysis we include only those banks that have both available offered and imputed interest rates. Our approach allows us to compare the differences in standard errors while the estimates are based on the same exact number of observations for the same banks in the same quarters.

Finally, a number of studies examine the effects that the presence of multimarket (i.e., larger, regional or national banks) may have on local market competition, specifically the impact on small singlemarket banks (community or rural banks). Cohen and Mazzeo (forthcoming) assess competition among financial institutions in rural markets by differentiating among different types institutions (single-market banks, multimarket banks and thrifts). The authors find that heterogeneous institution types affect competition and profitability and conclude that analysis of market concentration should address this differentiation. Berger, Dick, Goldberg, and White (forthcoming) find similar results. These authors

¹⁴ Savings accounts were not analyzed in Prager and Hannan (1998).

examine how competition from large, multimarket banks affects the performance of small, single-market banks in the same markets and how that competition changed over time. The authors find that technical progress enabled large multimarket banks to compete more effectively against small single-market banks in the 1990s. This finding did not hold with large banks that remained in a single market. Results of these studies provide strong support for conducting robustness checks on a sample of single-market banks, rather than examining both multimarket and single market banks together.

4. Empirical Analysis

We conduct three sets of tests to examine whether imputed deposit rates are representative of the deposit rates that banks offer to customers. First, we examine the distributions for the imputed and offered rates for different types of deposit accounts. Second, we test the null hypothesis that imputed interest rates are unbiased estimates of rates that banks offer by regressing offered deposit rates on their imputed counterparts in cross-sectional OLS regressions. In this step, we also examine whether winsorizing or truncating helps reduce any statistical bias that may exist. Finally, to examine whether observed statistical differences have any impact on empirical estimations, we estimate separate SCP regressions where the dependent variable is first the offered deposit rate and then the imputed deposit rate.

4.1 Alternative measures of imputed and offered rates

The MSSD is a monthly survey. Thus, we use the offered rates from last survey of the quarter (where R_MSSD_END denotes the MSSD rate at the quarter's end) and an average of the monthly surveyed deposit rates during the quarter (R_MSSD_AVE, where AVE denotes quarterly average of the MSSD rate). Included in our MSSD dataset are offered interest rates from 1987 to 1994 for three types of accounts: NOW, MMDA and savings accounts.

The two Call Report imputed rates that we use in our analysis are (1) the ratio of the end-ofquarter deposit expense (for each account type) divided by the end-of-quarter stock of deposits for the corresponding account type and quarter (R_CR_END) and (2) the ratio of the end-of-quarter deposit expense (for each account type) divided by the average stock of deposits over the quarter (R_CR_AVE).

4.2. Descriptive Statistics and Univariate Tests

In Table 1 Panel A we provide detailed distribution information on all of the interest rate series that we use in our analysis. The imputed rates obtained from the Call Report contain extreme outliers. Maxima and minima presented in Panel 1A. MSSD survey rates also contain interest rates that are too small or too high, suggesting that the survey data, too, contain outliers.

One method to minimize the effect of such extreme observations on the empirical analysis is to winsorize the variable in question by assigning the tail observations to a specified percentile of the data. We assign the observed values below the 1^{st} and above 99^{th} (5^{th} and 95^{th}) percentiles of the distribution, to the values of the 1^{st} and 99^{th} (5^{th} and 95^{th}) percentiles, respectively. Alternatively, one could truncate the series by dropping observations that are lower or higher than a given threshold at either end of the distribution. Each of these methods, however, has its weaknesses. While both methods preserve the central tendency of the distribution for the variable in question, truncation throws out valuable observations whereas winsorization assigns arbitrary values to observations that fall outside of the set threshold.

Panel 1A suggests that the large discrepancies observed in the tails of the distributions between R_MSSD and R_CR series are attenuated when we move to the 5th and 95th percentiles of series' distributions. The 25th, 50th and 75th quartile observations show relatively smaller differences. We, therefore, work with data series that are winsorized or truncated at 1^{st} –99th or 5th–95th percentiles. The tests that we conduct provide guidance whether winsorize or truncate and at what level. While higher levels of winsorization or truncation are certainly possible, we do not consider them here as it would lead to the loss of valuable data points and are unlikely to be adopted by researchers.

In Panel 1B, we test the statistical significance of the observed differences that we observe in Panel 1A for selected pairs of survey and imputed rate series that are winsorized at the 1st–99th percentiles

of their distributions (first three columns of Panel 1B) or truncated at the 5th-95th percentiles (last three columns). The t-tests of the equality of the means are rejected in all of the cases at the 1 percent level (except for the truncated savings account rates rejected at the 5 percent level). The equality of the variances (the variance ratio test) is also rejected at the one percent level for all six pairs of series that we examine. The rank-order correlations range between 0.51 to 0.89, and are the lowest for savings account series and the highest for the NOW account series. These results suggest that, even though the survey and imputed rate series that we examine are highly positively correlated, they may incorporate important differences that might lead to imputed rates being biased estimates of survey rates.

The results of Table 1 provide evidence of statistically significant differences between the offered rates obtained from MSSD data and the corresponding imputed rates obtained from the Call Report data (for the same set of banks in the same quarters).

There are a number of reasons why the imputed rates would not be good estimates of the offered rates. The imputed rates, a ratio of the interest expense and the stock of deposits, are affected by error in either component. Such errors include data entry errors, rounding issues, and reporting errors which may distort the imputed rate. Data entry errors are uncommon, but do exist. Such errors include: entries with an incorrect number of zeros, and general typos (reporting the wrong numbers or putting them in the wrong item number). All dollar amounts entered into the Call Reports are rounded to the nearest \$1,000. Thus, rounding issues are of greater concern for smaller banks which may end up having less accurate imputed rates. A number of reporting the interest expense, for example, banks are instructed to deduct from gross interest expense any penalties for early withdrawals or portions of such penalties that represent the forfeiture of interest accrued or paid to the date of withdrawal. This accounting requirement would reduce the amount of net interest expense reported by any bank that had early withdrawals.

Next, we investigate to what degree researchers of financial institutions should consider winsorization or truncation of imputed rates obtained from bank financial statements. Following this, we examine whether these observed differences generate biases of which researchers should be aware.

4.3. Winsorization versus Truncation of the Offered and Imputed Deposit Rates

These outliers, attributed to reporting errors, rounding errors, and mismeasurement, are substantial. Specifically, the MSSD-based MMDA offered rates range between 0.01% and 15.48%, whereas the corresponding raw imputed rates range between -60.53% and 216.05%. Similar outliers are observed for all of the offered and imputed deposit rate series in our samples.

To examine the effects of different levels of winsorization and truncation, we regress the winsorized (truncated) offered rate series on winsorized (truncated) imputed rate series with the same level of winsorization or truncation:

$$\mathbf{R}_{MSSD}_{i,t} = \alpha + \beta \ \mathbf{R}_{CR}_{i,t} + \varepsilon_{i,t}, \tag{1}$$

where R_MSSD_{i,t} is the offered rate collected from the MSSD survey, and R_CR_{i,t} is the corresponding Call Report imputed interest rate. We use $1^{st}-99^{th}$ and $5^{th}-95^{th}$ percentile thresholds when winsorizing or truncating. In regression (1), if imputed rates are unbiased estimates of the offered rates, the null hypothesis requires that $\alpha=0$ and $\beta=1$.¹⁵

The OLS regression estimates of equation (1) for NOW accounts are presented in Table 2.¹⁶ The left-hand-side of Table 2 presents results using the quarter-end rate for the survey data (R_MSSD_CR) and imputed rates based on the stock of NOW accounts at the end of the quarter (R_CR_END). The null hypothesis (H₀: α =0 and β =1) is rejected in Table 2. The F-test results are provided at the bottom of the table. When using the imputed rates winsorized at the 1st-99th percentile of their distribution, we obtain estimates of $\hat{\alpha}$ =0.0135 and $\hat{\beta}$ =0.6594. Truncating at the same threshold

¹⁵ This null hypothesis is in fact a joint-hypothesis since it presumes that offered transaction rates are the "true" interest rates in which researchers are interested.

¹⁶ We repeat the analysis using MMDA and savings account rates, but to conserve space, we do not to report these results.

level, we obtain estimates of $\hat{\alpha} = 0.0094$ and $\hat{\beta} = 0.7644$. Moving the threshold to the 5th-95th percentiles and winsorizing the series yields $\hat{\alpha} = 0.0064$ and $\hat{\beta} = 0.8473$, whereas truncating at the latter level yields $\hat{\alpha} = 0.0040$ and $\hat{\beta} = 0.9183$. The regression R²s systematically increase from 0.6333 to 0.8088. These results indicate that truncation at the higher threshold (5th-95th percentiles of the imputed rate distribution) provides a better fit than the other options tested. However, the null hypothesis of no bias (H₀: $\alpha=0$ and $\beta=1$) is rejected for all four regressions.

In the right-hand side of Table 2, we repeat the above analysis by regressing the average of rates surveyed in a given quarter (R_MSSD_AVE) on imputed rates that were calculated based on the average stock of deposits during the same quarter (R_CR_AVE). The results show that the bias is reduced in all of the regressions when compared with their counterparts on the left-hand side of Table 2. The coefficient estimates for the series winsorized at the 1st-99th percentiles are now $\hat{\alpha} = 0.0073$ and $\hat{\beta} = 0.8343$ with an R² of 0.7895. The coefficient estimates for the series for the series for the series truncated at the 5th-95th percentiles are $\hat{\alpha} = 0.0026$ and $\hat{\beta} = 0.9627$ with an R² of 0.8857. Despite a lower $\hat{\alpha}$ and a higher $\hat{\beta}$, the null hypothesis of no bias is still rejected in all of the regressions.

We conclude that survey rates and imputed rates truncated at the 5th and 95th percentiles of their distribution show the least amount of statistical bias. In the next section, we repeat this analysis and check the robustness of our results for rates on other account types.

4.4. Robustness Checks

In this section, we examine first whether the results observed for NOW accounts in Table 2 also hold for MMDA and savings accounts. We find that they do not. Columns 1 and 2 of Panel 3A repeat columns 4 and 8 of Table 2, respectively, for NOW accounts, while columns 3 and 4 of Panel 3A present the results for MMDA accounts.¹⁷ A sharp increase in the bias appears with regard to the MMDA rates results. As column 3 shows, regressing R_MSSD_END on R_CR_END for MMDA accounts yields $\hat{\alpha} = 0.0280$ and $\hat{\beta} = 0.4863$ with an R² of 0.3408 (compared to $\hat{\alpha} = 0.0038$ and $\hat{\beta} = 0.9244$ and an R² of 0.8107 for the comparable NOW account rates in the first column). Regressing R_MSSD_AVE on R_CR_AVE (column 4) yields $\hat{\alpha} = 0.0227$ and $\hat{\beta} = 0.5832$ with an R² of 0.4622. The observed bias is higher yet in the case of savings account rates (column 6); where $\hat{\alpha} = 0.0325$, $\hat{\beta} = 0.3689$ and R²=0.2964. Not surprisingly, the null hypothesis of no bias is strongly rejected in all of the cases in Panel 3A. When we repeat our bias regressions with the interest rate variables winsorized or truncated at 1st and 99th percentiles, we find larger bias (we do not report these results to conserve space).

Given the discrepancy in the bias tests between the NOW accounts versus the MMDA and savings accounts, we examine whether the observed differences hold in different subsamples. Specifically, we explore whether these discrepancies hold across (i) multimarket versus single-market banks (which may potentially explain the observed differences if NOW account rates have less dispersion than the MMDA or savings account rates for multimarket banks) and (ii) urban (MSA) versus rural banks. The results presented in panels 3B and 3C show that the results observed in Panel 3A for the overall sample also hold for the above-mentioned subsamples. NOW account rates show the least amount of bias.

In Panel 3B, we re-estimate regression (1) for multimarket and single-market banks with the series that show the least amount of bias (rates based on averages truncated at the 5th and 95th percentiles). The results for multimarket banks are very similar to those for single-market banks. For example, multimarket bank NOW account rates (column 1 of Panel 3B) yield estimates of $\hat{\alpha} = 0.0024$

¹⁷ The number of observations for the NOW account regressions differs slightly between Tables 2 and 3. We restrict the regressions in Table 3 (for all account types) to contain observations for which the end of quarter rates and average of the quarter rates are available for each bank. This allows better comparison of the bias inherent in either end or average rates for each account type.

and $\hat{\beta} = 0.9752$, with R²=0.8755. The single-market bank NOW account rates (column 4) yield $\hat{\alpha} = 0.0027$, $\hat{\beta} = 0.9599$ and R²=0.8982. Similar results hold for the MSSD and savings account rates.

We also examine whether the bias differs between urban (MSA) versus rural banks, two subsamples that researchers typically examine separately due to the differences in these markets' characteristics. In results presented in Panel 3C, we observe that the amount of bias (always statistically significant) remains about the same for banks operating in these two different types of markets for NOW and MMDA accounts. In the case of savings accounts, the bias for rates reported by urban banks is larger than the bias associated with the rates reported by rural banks.

These results suggest that there are no major discrepancies in the way imputed rates relate to survey rates when one compares the multi- versus single-market banks or urban versus rural banks. However, the bias is *statistically* significant for all rates considered in the different subsamples that we examined. What holds across these tests is that the observed bias is the smallest for NOW accounts, much larger for MMDAs, and even larger for savings accounts.

A number of reasons exist as to why the observed bias may differ across account types. The larger biases observed for MMDA and savings accounts may be due to the fact that banks are more likely to offer rate schedules tiered by deposit amount for MMDA and savings accounts than for NOW accounts. If so, then the offered schedules for MMDA and savings accounts would incorporate more dispersion than those for NOW accounts and mode of the distribution collected by the MSSD may be less representative of the distribution of rates for MMDA and savings accounts than it is for NOW accounts.

Another possible source of discrepancy is the minimum amount that is required for the payment of interest. Banks typically impose such minimum thresholds, and these are likely to be higher for MMDA and savings accounts than they are for NOW accounts. If so, MMDA and savings accounts may carry proportionally larger number of accounts that carry non-interest earning balances because they fell below the interest-earning threshold required by the bank. Despite these data errors and issues, the existence of statistical bias need not result in economically different inferences when imputed rates are used instead of offered rates. We test this conjecture next.

4.5. Imputed Deposit Rates versus Offered Deposit Rates in SCP Analysis

In this section, we compare the results of deposit price-market concentration regressions to determine whether using imputed interest rates instead of the offered interest rates leads to different conclusions in studies of the relationship between market concentration and prices (or tests of the SCP Hypothesis). While the scope of this test is limited to a specific case, it is relevant because deposit price information has been commonly used in SCP analysis. We conduct regressions based on equation (2) below and present the results in Table 4:

$$\mathbf{R}_{i,t} = a + b_1 H H I_{i,t} + b_2 M S A_{i,t} + b_3 H H I_{i,t} \times M S A_{i,t} + b_4 P I P C_{i,t} + \sum_{t=1}^{T} c_t D_t + e_{i,t}, \qquad (2)$$

where $R_{i,t}$ is the deposit-account specific offered or imputed interest rate (R_MSSD or R_CR , respectively), HHI is the Herfindahl-Hirshman Index of market concentration (calculated for each market once a year due to the availability of Summary of Deposits data), MSA is dummy variable that equals 1 for metropolitan markets and 0 otherwise, and PIPC is the personal income per capita in the market in question (in constant beginning-of-sample-period dollars, scaled by \$10,000). Deposit markets are defined as the largest of a county, a Metropolitan Statistical Area (MSA) or a Consolidated-MSA (CMSA). We interact the HHI variable with the MSA "dummy" to capture the potential differences in market concentration that may prevail in rural and urban markets. This interaction captures the marginal effect of an increase in concentration given that market concentration tends to be higher in rural markets (DeYoung, Goldberg and White, 1999). Finally, we account for time variation in deposit rates by including time "dummy" variables (D_i) for each quarter except the first.

We first present the results using the series that exhibited the highest amount of bias in Table 2, the offered rates from the last survey of the quarter (R_MSSD_END) and the corresponding imputed rate

(R_CR_END), both winsorized at the 1st and 99th percentiles of their distributions. The results of the SCP regressions using the MSSD data are presented in Panel 4A. The coefficient estimate for HHI is -0.0033 (statistically significant at the 1 percent level) when the NOW account *offered* rate (R_MSSD_END) is used as the dependent variable. This result is consistent with previous studies. When the NOW account imputed rate (R_CR_END) is used, however, the coefficient for HHI (-0.0035) has the expected sign, but is not statistically significant. The HHI interacted with MSA is positive and statistically significant in two of the offered rate regressions (the NOW account and savings account rates), but is not significant in any of the three imputed rate regressions. A positive sign on the HHI*MSA coefficient suggests that an increase in market concentration in urban (MSA) markets results in marginally lower deposit rates than in a non-MSA market.

While the coefficient estimates are similar, the standard errors are larger when Call Report imputed rates are used as the dependent variable. Indeed, all of the standard errors in column 2 (where R_CR_END is the dependent variable) are larger than their counterparts in column 1 (where R_MSSD_END is the dependent variable). This is not surprising, as the Call Report rates (the R_CR) presumably incorporate larger measurement error; and hence larger standard errors. Importantly, the measurement error in the dependent variable does not lead to a bias or inconsistent estimates but to less efficient estimates.

The results for the MMDA accounts (columns 3 and 4) and savings accounts (columns 5 and 6) are similar; we find variation in the coefficient estimates between offered (R_MSSD) and imputed (R_CR) rates, with the standard errors being larger in the case of the latter. The coefficient on HHI is not statistically significant for regressions using MMDA rates.

Results of SCP regressions for the series that exhibit the least amount of bias are presented in Panel B of Table 4. This panel contains the average of rates surveyed over the quarter (R_MSSD_AVE) versus imputed rates based on average deposits during the quarter (R_CR_AVE) and truncated at the 5th and 95th percentiles of their distributions. In general, the results using the two data sets are more similar in this panel than in Panel A. The coefficient estimates and standard errors are comparable between the

offered and imputed rates, and, consequentially, the signs and statistical significance of the coefficients would, in all cases but one, lead to like policy conclusions. A quick comparison of the two panels of Table 4 yields the following observations: First, in Panel 4B the discrepancy across coefficient estimates in columns 1 and 2 is much smaller, as are the standard deviations for the coefficient estimates. For example, in column 1 the coefficient estimate for HHI is equal to -0.0019 (with a standard error of 0.0013), in column 2 the corresponding coefficient estimate is -0.0012 (with a standard error of 0.0014). Again, neither of these coefficients is statistically significant. As in Panel A, the observed discrepancies for the same coefficient estimates are larger in the other columns as MMDA and savings account rate series exhibit larger biases. Even in those cases, however, the coefficient estimates in the SCP regressions for offered and imputed rates are similar and exhibit the same signs and significance levels; the coefficient on HHI is not significant in any of the regressions, but the coefficient on HHI*MSA is statistically significant in the regressions using the NOW imputed or offered rates. We find one notable difference in this set of regressions: HHI*MSA is significant in the regression using the savings account offered rate, but not the savings account imputed rate.

Based on these results, we recommend that researchers use the imputed series based on the average stock of deposits during the quarter and truncate the obtained variable at the 5th and 95th percentiles of its distribution. This choice would lead to coefficient estimates that are qualitatively similar to those that obtained from survey data, with the caveat that the former would have larger standard errors due to measurement error.¹⁸

As Prager and Hannan (1999) note, imputed interest rates are noisy estimates of offered (transaction) deposit rates. However, our work shows the importance of truncation and suggests that the

¹⁸ While these findings and conclusions seemingly contradict Prager and Hannan (1999), a number of discrepancies between the Prager and Hannan (1999) study and our study exist. First, Prager and Hannan (1999) estimate a model of price *changes* that compares pre- and post-merger prices whereas we estimate a SCP model in the price *levels*. Second, Prager and Hannan (1999) estimate the same model using *monthly* MSSD offered rates and *quarterly* Call Report imputed rates for the same time horizons. Thus the offered rate regressions would have a higher number of observations, and hence lower standard errors, than the regressions in which quarterly Call Report imputed rates are used. Conversely, our comparisons of offered versus imputed rates use the same number of observations in each of these regressions.

coefficient estimates obtained using the imputed rates (properly constructed and truncated) leads to analogous policy conclusions.

5. Summary and Conclusions

In this paper, we provide a systematic analysis of the bias that imputed interest rates may introduce when used as estimates of offered interest rates. Our results provide strong evidence of statistical bias. Imputed interest rates tend to underestimate the true interest rates offered to customers for the same account type at the same bank. We further examine whether this observed bias leads to economically different results in analyses that researchers conduct. Reassuringly, we find that this is not the case; typically the coefficient estimates have the same signs, same levels of statistical significance, and are of the same order of magnitude irrespective of whether imputed or offered rates are used. This suggests that imputed rates are suitable estimates of offered (transaction) rates when conducting empirical research. An important implication of our research is that empirical analyses of bank price data need not remain limited to survey data as imputed rates from the Call Reports are available for the population of U.S. banks.

References

- Adams, Robert M., Kenneth P. Brevoort and Elizabeth K. Kiser. (forthcoming). "Who Competes with Whom? The Case of Depository Institutions." Journal of Industrial Economics
- Berger, Allen N., Astrid A. Dick, Lawrence G. Goldberg and Lawrence J. White. (forthcoming). "Competition from Large, Multimarket Firms and the Performance of Small, Single-Market Firms: Evidence form the Banking Industry," *Journal of Money, Credit, and Banking*.
- Berger, Allen N. and Timothy H. Hannan. (1989). "The Price-Concentration Relationship In Banking," *Review of Economics and Statistics* 71(2): 291-299.
- Berger, Allen N. and Timothy H. Hannan. (1992). "The Price-Concentration Relationship in Banking: A Reply, *Review of Economics and Statistics*, 74 (February): 376-379.
- Brewer, Elijah III and William E. Jackson III. (2004) "The 'Risk-Adjusted' Price-Concentration Relationship in Banking." Federal Reserve Bank of Atlanta Working Paper 2004-35.
- Calem, Paul S. and Gerald A. Carlino. (1991). "The Concentration/Conduct Relationship in Bank Deposit Markets," *Review of Economics and Statistics* 73(2): 268-276.
- Calem, Paul S. and Leonard I. Nakamura. (1998). "Branch Banking and the Geography of Bank Pricing," *Review of Economics and Statistics* 80(4): 600-610.
- Cohen, Andrew and Michael J. Masseo. (2005). "Investment Strategies and Market Structure: An Empirical Analysis of Bank Branching Decisions." Working Paper.
- Cohen, Andrew and Michael J. Masseo. (forthcoming). "Market Structure and Competition Among Retail Depository Institutions." *Review of Economics and Statistics*.
- DeYoung, Goldberg, and White. (1999). "Youth, Adolescence and Maturity of Banks: Credit Availability to Small Business in an Era of Banking Consolidation," *Journal of Banking & Finance 23:* 463-492.
- Dick, Astrid. (2002). "Demand Estimation and Consumer Welfare in the Banking Industry," Board of Governors of the Federal Reserve System Working Paper 2002-58.
- Gilbert, R. Alton. (1984) "Bank Market Structure and Competition: A Survey," *Journal of Money, Credit, and Banking* 16 (November): 617-45.
- Hannan, Timothy H. (1991). "Bank Commercial Loan Markets and the Role of Market Structure: Evidence from Surveys of Commercial Lending", *Journal of Banking and Finance* 15: 133-149.
- Hannan, Timothy H. (1997). "Market Share Inequality, The Number Of Competitors, And The HHI: An Examination Of Bank Pricing," *Review of Industrial Organization* 12(1): 23-35.
- Hannan, Timothy H. (2002). "The Impact of Credit Unions on the Rates Offered For Retail Deposits by Banks and Thrift Institutions," Board of Governors of the Federal Reserve System Working Paper 2002-06.

- Hannan, Timothy H. (2006). "Retail Deposit Fees and Multimarket Banking," *Journal of Banking and Finance 30*(9): 2561-78.
- Hannan, Timothy H. and Allen N. Berger. (1991). "The Rigidity of Prices: Evidence from the Banking Industry," *American Economic Review* 81(4): 938-945.
- Hannan, Timothy H. and Robin A. Prager. (2006). "Multimarket Bank Pricing: An Empirical Investigation of Bank Deposit Interest Rates," *Journal of Economics and Business* 58(3):256-72.
- Hannan, Timothy H. and Robin A. Prager. (2004). "The Competitive Implications of Multimarket Bank Branching," *Journal of Banking and Finance* 28(8): 1889-1914.
- Heitfield, E.A. (1999). "What Do Interest Rate Data Say About the Geography of Retail Banking Markets?" *The Antitrust Bulletin* 44(2) 333-347.
- Heitfield, Erik A. and Robin A. Prager. (2002). "The Geographic Scope of Retail Deposit Markets." Board of Governors of the Federal Reserve System Working Paper 2002-49.
- Jackson, William E. III. (1992). "The Price-Concentration Relationship in Banking: A Comment," *Review of Economics and Statistics* 74 (February): 373-376.
- Kiser, Elizabeth K. (2004). "Modeling the Whole Firm: The Effect of Multiple Inputs and Financial Intermediation on Bank Deposit Rates." Board of Governors of the Federal Reserve System Working Paper 2004-07.
- Neumark, David, and Steven A. Sharpe. (1992). "Market Structure and the Nature of Price Rigidity: Evidence from the Market for Consumer Deposits." *Quarterly Journal of Economics* 107 (May): 657-80.
- Prager, Robin. A. and Timothy H. Hannan. (1998). "Do Substantially Horizontal Mergers Generate Significant Price Effects? Evidence from the Banking Industry," *The Journal of Industrial Economics* 46(4): 433-452.
- Prager, Robin A. and Timothy H. Hannan. (1999). "The Price Effects of Horizontal Mergers in the Banking Industry: Some Further Evidence," Working Paper.
- Radecki, Lawrence J. (1998). "The Expanding Geographic Reach of Retail Banking Markets," *Federal Reserve Bank of New York Economic Policy Review* (June) 15-33.
- Rhoades, Stephen A., and Roger D. Rutz, (1982). "Market Power and Firm Risk: A Test of the 'Quiet Life' Hypothesis," *Journal of Monetary Economics* 9: 73-85.
- Rosen, Richard J. (2002). "What Goes Up Must Come Down? Asymmetries and Persistence in Bank Deposit Rates." Journal of Financial Services Research 21 (3): 173-193.
- Rosen, Richard J. (2002). "Banking Market Conditions and Deposit Interest Rates." Federal Reserve Bank of Chicago Working Paper 2003-19.

Table 1. Summary Statistics and Univariate Tests

Panel A presents distributional information on Monthly Survey of Selected Deposits (MSSD) survey rates and Call Report (CR) imputed rates. END refers to MSSD series based on the last survey of quarter and CR series based on the stock of deposits at the end of the same quarter. AVE refers to MSSD series based on the average of surveyed rates during the quarter and CR series based on the average stock of deposits during the quarter. Panel B presents tests of equality of the means (t-test with H₀: mean_X – mean_Y = 0), tests of equality of the variances (F-test with H₀: var_X ÷ var_Y = 1). And rank-order correlations (ρ) for selected pairs of series from Panel A. * and ** denote statistical significance at the 5% and 1% levels, respectively.

Panel 1A					1^{st}	5 th	25 th		75 th	95 th	99 th	
			Ν	Minimum	percentile	percentile	percentile	Median	percentile	percentile	percentile	Maximum
NOW	1987Q2-	R_MSSD_END	14,445	0.0001	0.0120	0.0151	0.0275	0.0470	0.0505	0.0550	0.0608	0.1548
Accounts	1994Q2	R_CR_END	14,445	-0.8296	0.0098	0.0155	0.0275	0.0439	0.0502	0.0617	0.1027	1.3333
		R_MSSD_AVE	14,445	0.0014	0.0122	0.0161	0.0285	0.0470	0.0509	0.0548	0.0608	0.1528
		R_CR_AVE	14,445	-1.9801	0.0097	0.0155	0.0273	0.0443	0.0499	0.0570	0.0782	1.0000
MMDA	1987Q1-	R_MSSD_END	8,339	0.0350	0.0450	0.0495	0.0525	0.0550	0.0600	0.0710	0.0800	0.1150
	1991Q2	R_CR_END	8,339	-0.0771	0.0245	0.0466	0.0534	0.0576	0.0635	0.0747	0.0975	1.3333
		R_MSSD_AVE	8,339	0.0350	0.0460	0.0494	0.0525	0.0553	0.0600	0.0708	0.0794	0.1050
		R_CR_AVE	8,339	-0.0967	0.0262	0.0479	0.0533	0.0570	0.0629	0.0741	0.0925	1.7770
Savings	1987Q1-	R_MSSD_END	8,442	0.0400	0.0425	0.0450	0.0500	0.0510	0.0550	0.0575	0.0737	0.0895
Accounts	1991Q2	R_CR_END	8,442	-0.8703	0.0223	0.0415	0.0485	0.0512	0.0547	0.0691	0.1157	0.4727
		R_MSSD_AVE	8,442	0.0386	0.0425	0.0450	0.0500	0.0516	0.0550	0.0576	0.0733	0.0892
		R_CR_AVE	8,442	-0.8387	0.0224	0.0431	0.0492	0.0514	0.0549	0.0690	0.1186	1.1104

Panel 1B

Rates Winsorized at 1st & 99th Percentiles

Rates *Truncated* at 5th & 95th Percentiles

		R_MSSD_END	R_CR_END	tests	-	R_MSSD_AVE	R_CR_AVE	tests	
NOW	Mean	0.0404	0.0408	-2.44	**	0.0408	0.0396	7.94	**
Accounts	Standard Deviation	0.0134	0.0161	0.69	**	0.0118	0.0115	1.05	**
	Rank-order correlation (ρ)			0.82	**			0.89	**
MMDA	Mean	0.0571	0.0588	-13.02	**	0.0562	0.0574	-13.64	**
	Standard Deviation	0.0068	0.0099	0.47	**	0.0046	0.0053	0.74	**
	Rank-order correlation (ρ)			0.59	**			0.69	**
Savings	Mean	0.0520	0.0527	-5.63	**	0.0515	0.0514	-1.80	*
Accounts	Standard Deviation	0.0044	0.0113	0.15	**	0.0027	0.0040	0.46	**
	Rank-order correlation (ρ)			0.51	**			0.64	**

Table 2. Winsorize or Truncate the Deposit Rates?

This table presents the results of the following OLS regression for unbalanced panels of U.S. commercial banks:

$$R_{MSSD_{i,t}} = \alpha + \beta R_{CR_{i,t}} + \varepsilon_{i,t}$$
(1)

The dependent variable is the interest rate that the bank has most commonly offered for NOW accounts and reported in the last survey of the quarter. The explanatory variable is the imputed interest rate for NOW accounts obtained from the Call Reports for the same bank in the same quarter. All equations are estimated using OLS with Huber-White standard errors (reported in parentheses below coefficient estimates) for clustering across banks. F-statistics are reported for the joint-null hypothesis that $\alpha=0$ and $\beta=1$. * and ** denote significance at the 5% and 1% levels, respectively.

NOW Account Rates																
	R_MSS	D_I	$\text{END}_{i,t} = 0$	α+	βR_CR_	ENI	$D_{i,t} + \mathcal{E}_{i,t}$		$\underline{\mathbf{R}}_{\mathrm{MSSD}} \underline{\mathbf{AVE}}_{i,t} = \alpha + \beta \underline{\mathbf{R}}_{\mathrm{CR}} \underline{\mathbf{AVE}}_{i,t} + \varepsilon_{i,t}$							
	R_MSSD winsorized at 1%		R_MSSD truncated at 1%		R_MSSD winsorized at 5%		R_MSSD truncated at 5%		R_MSSD winsorized at 1%		R_MSSD truncated at 1%		R_MSSD winsorized at 5%		R_MSSD truncated at 5%	
Constant	0.0135 (0.0009)	**	0.0094 (0.0008)	**	0.0064 (0.0005)	**	0.0040 (0.0003)	**	0.0073 (0.0005)	**	0.0044 (0.0003)	**	0.0044 (0.0004)	**	0.0026 (0.0003)	**
R_CR winsorized at 1%	0.6594 (0.0244)	**							0.8343 (0.0132)	**						
R_CR truncated at 1%			0.7644 (0.0212)	**							0.9083 (0.0087)	**				
R_CR winsorized at 5%					0.8473 (0.0112)	**							0.9158 (0.0084)	**		
R_CR truncated at 5%							0.9183 (0.0076)	**							0.9627 (0.0057)	**
$H_0: \alpha = 0 \& \beta = 1$	118.9	**	95.1	**	99.8	**	72.9	**	105.7	**	105.0	**	88.9	**	95.3	**
Bank-quarter obs. Number of banks R ²	14,446 735 0.6333		13,912 730 0.7175		14,446 735 0.7760		12,564 720 0.8088		14,973 737 0.7895		14,393 735 0.8542		14,973 737 0.8528		12,751 721 0.8857	
Regression F-stat	731.4	**	1296.5	**	5680.7	**	14702.3	**	4007.8	**	10981.0	**	12015.4	**	28312.8	**

Table 3. Tests of Statistical Bias

This table presents the results of the following OLS regression for unbalanced panels of U.S. commercial banks:

$$\mathbf{R}_{MSSD}_{i,t} = \alpha + \beta \ \mathbf{R}_{CR}_{i,t} + \varepsilon_{i,t}$$
(1)

The dependent variable is the interest rate that the bank has most commonly offered for a particular type of deposit account. We use either the rate from the <u>last</u> survey of the quarter or the <u>average</u> of rates in all surveys of the quarter. The explanatory variable is the imputed interest rate obtained from the Call Reports. We use either the stock of deposits as reported at the <u>end</u> of the quarter or the <u>average</u> stock of deposits through the quarter (as reported in Schedule K of Call Reports). All imputed rates are <u>truncated at 5%</u> of either side of their distributions. All equations are estimated using ordinary least squares (OLS) with Huber-White standard errors for clustering across banks. F-statistics are reported for the joint-null hypothesis that $\alpha=0$ and $\beta=1$. T-statistics appear in parentheses. * and ** denote significance at the 5% and 1% levels, respectively.

Panel 3A	(all rates truncated at 5 th & 95 th percentile of their distributions)											
	NOW Accounts				MMDA				Savings Accounts			
	R_MSSD_END		R_MSSD_AVE		R_MSSD_END		R_MSSD_AVE		R_MSSD_END		R_MSSD_AVE	:
Constant	0.0038 (0.0003)	**	0.0026 (0.0003)	**	0.0280 (0.0009)	**	0.0227 (0.0011)	**	0.0336 (0.0011)	**	0.0325 (0.0015)	**
R_CR_END	0.9244 (0.0077)	**			0.4863 (0.0169)	**			0.3490 (0.0205)	**		
R_CR_AVE			0.9641 (0.0058)	**			0.5832 (0.0196)	**			0.3689 (0.0280)	**
Η ₀ : α=0 & β=1	68.3	**	97.1	**	462.2	**	230.4	**	511.2	**	255.0	**
# of bank-quarter obs. # of banks R ²	11,967 714 0.8107	**	11,967 714 0.8858	**	6,646 619 0.3408	**	6,646 619 0.4622	**	6,914 614 0.2655	**	6,914 614 0.2964	**
Regression F-stat	14509.2	**	27475.3	4 A	828.0	ተቸ	884.5	Υ	289.8	**	173.2	ተቸ

	<u>Multimarket Banks vs. Single-Market Banks</u>																	
Panel 3B			R MSS	SD .	$AVE_{ii} = c$	ε + <i>L</i>	3 R CR	AV	$E_{i,i} + \varepsilon_{i,i}$									
		(011)	—	- nonte	i,i ad at tha 5 th (, 2. 05 ^t	h porcontilo	- of th	i,i <i>i,i</i>	tion	.)							
		(all I	ales ale liu	ncate	eu at tile 5 c	x 95	percentile	01 11		mons	5)							
		Mu	ltimarket E	Bank	S			Sing	le-Market	Banl	KS							
					Savings		Savings											
	NOW		MMDA		Accounts		NOW		MMDA		Accounts							
Constant	0.0024	**	0.0271	**	0.0363	**	0.0027	**	0.0282	**	0.0392	**						
	(0.0004)		(0.0022)		(0.0018)		(0.0004)		(0.0023)		(0.0028)							
R CR AVE	0.9756	**	0.5049	**	0.2901	**	0.9599	**	0.4901	**	0.2470	**						
	(0.0087)		(0.0379)		(0.0346)		(0.0086)		(0.0408)		(0.0534)							
$H_{a} \cdot \alpha = 0 \& \beta = 1$	55.4	**	112.9	**	210.0	**	41.1	**	87.5	**	100.9	**						
11 ₀ . u=0 & p=1	55.4		112.9		210.0		41.1		07.5		100.9							
# of bank-quarter obs.	6,265		3,537		3,574		4,563		2,746		2,861							
# of banks	394		322		318		348		290		284							
R ²	0.8755	ale ale	0.3940	ale ale	0.2255	ماد ماد	0.8982	باد باد	0.4010	ماد ماد	0.1848	ale ale						
Regression F-stat	12538.5	**	177.8	**	70.1	**	12441.7	**	144.4	**	21.4	<u> </u>						
			<u>l</u>	Jrba	<u>n (MSA) vs.</u>	Rur	al Market	Ban	ks									
Panel 3C			R_MSS	SD_{-}	$AVE_{it} = c$	ε + <i>μ</i>	3 R_CR_	AV	$E_{it} + \varepsilon_{it}$	R MSSD AVE: $= \alpha + \beta$ R CR AVE: $+ \varepsilon$.								
		(all rates are truncated at the 5 th & 05 th percentile of their distributions)																
	(an rates are truncated at the 5 & 55 percentile of their distributions)																	
		(all i	ates are tru	ncate	ed at the $5^{\text{th}} \delta$	& 95 ^t	^h percentile	of th	neir distribu	itions	5)							
	Urb	(all 1 an (1	rates are tru MSA) Mar	ncate ket F	ed at the 5 th &	& 95 ^t	^h percentile	of th Rura	neir distribu al Market	itions Bank	s) ss							
	Urb	<u>(all 1</u> an (1	rates are tru MSA) Mar	ncate ket F	ed at the 5 th & Banks Savings	& 95 ^t	^h percentile	of the	al Market	itions Bank	s) ss Savings							
	Urb NOW	<u>(all 1</u> an (1	rates are tru MSA) Mar MMDA	ncate ket E	ed at the 5 th & Banks Savings Accounts	& 95 ^t	^h percentile NOW	of th Rur :	neir distribu al Market 1 MMDA	itions Bank	s) Savings Accounts							
Constant	Urb NOW 0.0024	<u>(all 1</u> an (1	rates are tru MSA) Mar MMDA 0.0275	ncate ket E	ed at the 5 th & Banks Savings Accounts 0.0383	**	h percentile NOW 0.0030	of the Rura	neir distribu al Market 1 MMDA 0.0287	ations Bank	s) Savings Accounts 0.0303	**						
Constant	Urb NOW 0.0024 (0.0003)	<u>(all 1</u> an (! **	rates are tru MSA) Mar MMDA 0.0275 (0.0018)	ncate	ed at the 5 th & Banks Savings Accounts 0.0383 (0.0018)	** **	h percentile NOW 0.0030 (0.0007)	of the Rura	neir distribu al Market 1 <u>MMDA</u> 0.0287 (0.0038)	ntions Bank	s) Savings Accounts 0.0303 (0.0029)	**						
Constant R CR AVE	Urb NOW 0.0024 (0.0003) 0.9712	<u>(all 1</u> an (! **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002	ncate ket E **	ed at the 5 th & Banks Savings <u>Accounts</u> 0.0383 (0.0018) 0.2565	& 95 ^t	h percentile NOW 0.0030 (0.0007) 0.9509 0	<u>of th</u> Rur **	neir distribu al Market 1 MMDA 0.0287 (0.0038) 0.4737	Bank	s) Savings <u>Accounts</u> 0.0303 (0.0029) 0.4159	**						
Constant R_CR_AVE	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068)	<u>(all 1</u> an (! **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307)	ncate	ed at the 5 th & Banks Savings Accounts 0.0383 (0.0018) 0.2565 (0.0344)	**	^h percentile <u>NOW</u> 0.0030 (0.0007) 0.9509 (0.0156)	<u>of th</u> <u>Rur:</u> **	neir distribu al Market 1 MMDA 0.0287 (0.0038) 0.4737 (0.0678)	Bank **	Savings Accounts 0.0303 (0.0029) 0.4159 (0.0572)	**						
Constant R_CR_AVE	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068)	(all 1 an (! **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307)	<u>ncate</u> <u>ket F</u> **	ed at the 5 th & Banks Savings Accounts 0.0383 (0.0018) 0.2565 (0.0344)	** **	NOW 0.0030 (0.0007) 0.9509 (0.0156)	of th Rur: ** **	neir distribu al Market I MMDA 0.0287 (0.0038) 0.4737 (0.0678)	<u>Bank</u> **	s) Savings Accounts 0.0303 (0.0029) 0.4159 (0.0572)	**						
Constant R_CR_AVE H ₀ : α=0 & β=1	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068) 83.4	(all 1 an (1 ** **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307) 163.3	ncate ket E ** **	ed at the 5 th & Banks Savings <u>Accounts</u> 0.0383 (0.0018) 0.2565 (0.0344) 236.8	** **	NOW 0.0030 (0.0007) 0.9509 (0.0156) 13.4	of th Rur: ** **	Matrix Market MMDA 0.0287 (0.0038) 0.4737 0.0678) 33.8	Bank **	s) Savings <u>Accounts</u> 0.0303 (0.0029) 0.4159 (0.0572) 53.5	**						
Constant R_CR_AVE H ₀ : α=0 & β=1 # of bank-quarter obs	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068) 83.4 9.140	(all 1 an (1 ** **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307) 163.3 5.325	ncate ket E ** **	ed at the 5 th & Banks Savings Accounts 0.0383 (0.0018) 0.2565 (0.0344) 236.8 5.441	** **	NOW 0.0030 (0.0007) 0.9509 (0.0156) 13.4	of th Rur: ** **	Antic America Antice America Antice America Am	Bank ** **	s) Savings Accounts 0.0303 (0.0029) 0.4159 (0.0572) 53.5 994	**						
Constant R_CR_AVE H ₀ : α=0 & β=1 # of bank-quarter obs. # of banks	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068) 83.4 9,140 558	(all 1 an (f ** **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307) 163.3 5,325 479	<u>ncate</u> <u>ket F</u> ** **	ed at the 5 th & Banks Savings Accounts 0.0383 (0.0018) 0.2565 (0.0344) 236.8 5,441 474	** **	^h percentile <u>NOW</u> 0.0030 (0.0007) 0.9509 (0.0156) 13.4 1,688 106	<u>of th</u> <u>Rur</u> ** **	neir distribu al Market 1 MMDA 0.0287 (0.0038) 0.4737 (0.0678) 33.8 958 90	**	s) Savings Accounts 0.0303 (0.0029) 0.4159 (0.0572) 53.5 994 88	**						
Constant R_CR_AVE H ₀ : α=0 & β=1 # of bank-quarter obs. # of banks R ²	Urb NOW 0.0024 (0.0003) 0.9712 (0.0068) 83.4 9,140 558 0.8880	(all 1 an (! ** **	MSA) Mar MMDA 0.0275 (0.0018) 0.5002 (0.0307) 163.3 5,325 479 0.3926	<u>ncate</u> ket F ** **	ed at the 5 th & Banks Savings <u>Accounts</u> 0.0383 (0.0018) 0.2565 (0.0344) 236.8 5,441 474 0 1929	<u>k 95^t</u> ** **	NOW 0.0030 (0.0007) 0.9509 (0.0156) 13.4 1,688 106 0.8657	**	neir distribu al Market 1 MMDA 0.0287 (0.0038) 0.4737 (0.0678) 33.8 958 90 0.4113	tions Bank ** **	s) Savings Accounts 0.0303 (0.0029) 0.4159 (0.0572) 53.5 994 88 0 3241	***						

Table 3. Tests of Statistical Bias (continued)

Table 4. Imputed Vs. Offered Rates in Structure Conduct Performance Analysis

$$\mathbf{R}_{i,t} = a + b_1 H H I_{i,t} + b_2 M S A_{i,t} + b_3 H H I_{i,t} \times M S A_{i,t} + b_4 P I P C_{i,t} + \sum_{t=1}^{I} c_t D_t + e_{i,t}$$
(2)

The dependent variable is either (i) the MSSD offered interest rate (R_MSSD) or (ii) the Call Report imputed rate (R_CR). HHI is the Herfindahl-Hirshman Index of market concentration. MSA is a "dummy" variable for metropolitan markets. PIPC is the personal income per capita (in constantbeginning-of-sample-year dollars). OLS regressions control for clustering at the bank level using Huber-White standard errors. T-statistics appear in parentheses below the estimated coefficients. * and ** denote statistical significance at the 5% and 1% levels, respectively.

Panel 4A MSSD Offered Rates versus CR Imputed Rates in SCP Regressions

	l	NOW	V		Ν	IMD	A		Saving	gs Ac	counts	
	R_MSSD_END		R_CR_END		R_MSSD_END		R_CR_END		R_MSSD_END		R_CR_END	
Constant	0.0250 (0.0016)	**	0.0225 (0.0025)	**	0.0508 (0.0014)	**	0.0553 (0.0016)	**	0.0482 (0.0014)	**	0.0458 (0.0031)	**
ННІ	-0.0033 (0.0015)	*	-0.0035 (0.0027)		-0.0018 (0.0024)		-0.0015 (0.0026)		-0.0003 (0.0015)		0.0041 (0.0048)	
MSA	-0.0020 (0.0009)	*	-0.0035 (0.0016)	*	-0.0014 (0.0013)		-0.0007 (0.0015)		-0.0002 (0.0008)		0.0002 (0.0019)	
HHI×MSA	-0.0067 (0.0028)	*	-0.0042 (0.0045)		-0.0020 (0.0038)		-0.0034 (0.0047)		-0.0065 (0.0025)	**	-0.0087 (0.0061)	
PIPC (in \$10,000)	-0.0008 (0.0005)		0.0002 (0.0008)		0.0030 (0.0007)	**	0.0017 (0.0009)		0.0015 (0.0005)	**	0.0013 (0.0009)	
Quarter dummies	yes		yes		yes		yes		yes		yes	
# of bank-quarter obs.	10,956		10,956		6,348		6,348		6,343		6,343	
# of banks	589		589		511		511		511		511	
\mathbf{R}^2	0.8609		0.6213		0.2553		0.1742		0.0641		0.0268	
Regression F-stat	406.2	**	268.0	**	50.2	**	61.8	**	10.8	**	18.3	**

(all rates <u>winsorized</u> at 1st & 99th percentile of their distributions)

Table 4. Imputed Vs. Offered Rates in Structure Conduct Performance Analysis (continued)

The dependent variable is either (i) the MSSD offered interest rate (R_MSSD) or (ii) the Call Report imputed rate (R_CR). HHI is the Herfindahl-Hirshman Index of market concentration. MSA is a "dummy" variable for metropolitan markets. PIPC is the personal income per capita (in constantbeginning-of-sample-year dollars). OLS regressions control for clustering at the bank level using Huber-White standard errors. T-statistics appear in parentheses below the estimated coefficients. * and ** denote statistical significance at the 5% and 1% levels, respectively.

Panel 4B **MSSD Offered Rates versus CR Imputed Rates in SCP Regressions** (all rates truncated at 5th & 95th percentile of their distributions) NOW **MMDA Savings Accounts** R_MSSD_AVE R_MSSD_AVE R_MSSD_AVE R_CR_AVE **R_CR_AVE** R CR AVE 0.0255 0.0250 0.0476 0.0486 0.0496 0.0500 ** Constant ** ** ** ** (0.0013)(0.0015)(0.0013)(0.0014)(0.0010)(0.0012)HHI -0.0019-0.00120.0005 0.0008 -0.0004-0.0009(0.0013)(0.0014)(0.0021)(0.0020)(0.0015)(0.0015)MSA -0.0017 * -0.0018-0.00010.0002 -0.0002-0.0011(0.0008)(0.0009)(0.0009)(0.0009)(0.0007)(0.0008)** HHI×MSA -0.0057 * -0.0066-0.0029-0.0031-0.0053 * -0.0042(0.0024)(0.0025)(0.0030)(0.0031)(0.0021)(0.0026)** ** 0.0021 0.0017 **PIPC (in \$10,000)** -0.0003-0.00010.0006 0.0007 (0.0005)(0.0005)(0.0005)(0.0005)(0.0005)(0.0004)**Ouarter "dummies"** yes yes yes yes yes yes 9,763 9,763 5,746 5,746 5,887 5,887 # of bank-quarter obs. 579 # of banks 579 514 514 503 503 \mathbf{R}^2 0.8800 0.8215 0.2671 0.2607 0.0626 0.0354 ** ** ** ** ** 414.9 372.3 39.9 53.7 7.7 6.9 **Regression F-stat**

Working Paper Series

A series of research studies on regional economic issues relating to the Seventh Federal Reserve District, and on financial and economic topics.

A Proposal for Efficiently Resolving Out-of-the-Money Swap Positions at Large Insolvent Banks <i>George G. Kaufman</i>	WP-03-01
Depositor Liquidity and Loss-Sharing in Bank Failure Resolutions George G. Kaufman	WP-03-02
Subordinated Debt and Prompt Corrective Regulatory Action Douglas D. Evanoff and Larry D. Wall	WP-03-03
When is Inter-Transaction Time Informative? Craig Furfine	WP-03-04
Tenure Choice with Location Selection: The Case of Hispanic Neighborhoods in Chicago Maude Toussaint-Comeau and Sherrie L.W. Rhine	WP-03-05
Distinguishing Limited Commitment from Moral Hazard in Models of Growth with Inequality* Anna L. Paulson and Robert Townsend	WP-03-06
Resolving Large Complex Financial Organizations Robert R. Bliss	WP-03-07
The Case of the Missing Productivity Growth: Or, Does information technology explain why productivity accelerated in the United States but not the United Kingdom? <i>Susanto Basu, John G. Fernald, Nicholas Oulton and Sylaja Srinivasan</i>	WP-03-08
Inside-Outside Money Competition Ramon Marimon, Juan Pablo Nicolini and Pedro Teles	WP-03-09
The Importance of Check-Cashing Businesses to the Unbanked: Racial/Ethnic Differences William H. Greene, Sherrie L.W. Rhine and Maude Toussaint-Comeau	WP-03-10
A Firm's First Year Jaap H. Abbring and Jeffrey R. Campbell	WP-03-11
Market Size Matters Jeffrey R. Campbell and Hugo A. Hopenhayn	WP-03-12
The Cost of Business Cycles under Endogenous Growth Gadi Barlevy	WP-03-13
The Past, Present, and Probable Future for Community Banks Robert DeYoung, William C. Hunter and Gregory F. Udell	WP-03-14

Measuring Productivity Growth in Asia: Do Market Imperfections Matter? John Fernald and Brent Neiman	WP-03-15
Revised Estimates of Intergenerational Income Mobility in the United States Bhashkar Mazumder	WP-03-16
Product Market Evidence on the Employment Effects of the Minimum Wage Daniel Aaronson and Eric French	WP-03-17
Estimating Models of On-the-Job Search using Record Statistics Gadi Barlevy	WP-03-18
Banking Market Conditions and Deposit Interest Rates Richard J. Rosen	WP-03-19
Creating a National State Rainy Day Fund: A Modest Proposal to Improve Future State Fiscal Performance <i>Richard Mattoon</i>	WP-03-20
Managerial Incentive and Financial Contagion Sujit Chakravorti and Subir Lall	WP-03-21
Women and the Phillips Curve: Do Women's and Men's Labor Market Outcomes Differentially Affect Real Wage Growth and Inflation? <i>Katharine Anderson, Lisa Barrow and Kristin F. Butcher</i>	WP-03-22
Evaluating the Calvo Model of Sticky Prices Martin Eichenbaum and Jonas D.M. Fisher	WP-03-23
The Growing Importance of Family and Community: An Analysis of Changes in the Sibling Correlation in Earnings <i>Bhashkar Mazumder and David I. Levine</i>	WP-03-24
Should We Teach Old Dogs New Tricks? The Impact of Community College Retraining on Older Displaced Workers <i>Louis Jacobson, Robert J. LaLonde and Daniel Sullivan</i>	WP-03-25
Trade Deflection and Trade Depression Chad P. Brown and Meredith A. Crowley	WP-03-26
China and Emerging Asia: Comrades or Competitors? Alan G. Ahearne, John G. Fernald, Prakash Loungani and John W. Schindler	WP-03-27
International Business Cycles Under Fixed and Flexible Exchange Rate Regimes <i>Michael A. Kouparitsas</i>	WP-03-28
Firing Costs and Business Cycle Fluctuations Marcelo Veracierto	WP-03-29
Spatial Organization of Firms Yukako Ono	WP-03-30
Government Equity and Money: John Law's System in 1720 France <i>François R. Velde</i>	WP-03-31

Deregulation and the Relationship Between Bank CEO Compensation and Risk-Taking Elijah Brewer III, William Curt Hunter and William E. Jackson III	WP-03-32
Compatibility and Pricing with Indirect Network Effects: Evidence from ATMs Christopher R. Knittel and Victor Stango	WP-03-33
Self-Employment as an Alternative to Unemployment Ellen R. Rissman	WP-03-34
Where the Headquarters are – Evidence from Large Public Companies 1990-2000 <i>Tyler Diacon and Thomas H. Klier</i>	WP-03-35
Standing Facilities and Interbank Borrowing: Evidence from the Federal Reserve's New Discount Window <i>Craig Furfine</i>	WP-04-01
Netting, Financial Contracts, and Banks: The Economic Implications William J. Bergman, Robert R. Bliss, Christian A. Johnson and George G. Kaufman	WP-04-02
Real Effects of Bank Competition Nicola Cetorelli	WP-04-03
Finance as a Barrier To Entry: Bank Competition and Industry Structure in Local U.S. Markets? Nicola Cetorelli and Philip E. Strahan	WP-04-04
The Dynamics of Work and Debt Jeffrey R. Campbell and Zvi Hercowitz	WP-04-05
Fiscal Policy in the Aftermath of 9/11 Jonas Fisher and Martin Eichenbaum	WP-04-06
Merger Momentum and Investor Sentiment: The Stock Market Reaction To Merger Announcements <i>Richard J. Rosen</i>	WP-04-07
Earnings Inequality and the Business Cycle Gadi Barlevy and Daniel Tsiddon	WP-04-08
Platform Competition in Two-Sided Markets: The Case of Payment Networks Sujit Chakravorti and Roberto Roson	WP-04-09
Nominal Debt as a Burden on Monetary Policy Javier Díaz-Giménez, Giorgia Giovannetti, Ramon Marimon, and Pedro Teles	WP-04-10
On the Timing of Innovation in Stochastic Schumpeterian Growth Models <i>Gadi Barlevy</i>	WP-04-11
Policy Externalities: How US Antidumping Affects Japanese Exports to the EU Chad P. Bown and Meredith A. Crowley	WP-04-12
Sibling Similarities, Differences and Economic Inequality Bhashkar Mazumder	WP-04-13

Determinants of Business Cycle Comovement: A Robust Analysis Marianne Baxter and Michael A. Kouparitsas	WP-04-14
The Occupational Assimilation of Hispanics in the U.S.: Evidence from Panel Data <i>Maude Toussaint-Comeau</i>	WP-04-15
Reading, Writing, and Raisinets ¹ : Are School Finances Contributing to Children's Obesity? <i>Patricia M. Anderson and Kristin F. Butcher</i>	WP-04-16
Learning by Observing: Information Spillovers in the Execution and Valuation of Commercial Bank M&As <i>Gayle DeLong and Robert DeYoung</i>	WP-04-17
Prospects for Immigrant-Native Wealth Assimilation: Evidence from Financial Market Participation Una Okonkwo Osili and Anna Paulson	WP-04-18
Individuals and Institutions: Evidence from International Migrants in the U.S. <i>Una Okonkwo Osili and Anna Paulson</i>	WP-04-19
Are Technology Improvements Contractionary? Susanto Basu, John Fernald and Miles Kimball	WP-04-20
The Minimum Wage, Restaurant Prices and Labor Market Structure Daniel Aaronson, Eric French and James MacDonald	WP-04-21
Betcha can't acquire just one: merger programs and compensation <i>Richard J. Rosen</i>	WP-04-22
Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked <i>Lisa Barrow and Kristin F. Butcher</i>	WP-04-23
The Role of Collateralized Household Debt in Macroeconomic Stabilization <i>Jeffrey R. Campbell and Zvi Hercowitz</i>	WP-04-24
Advertising and Pricing at Multiple-Output Firms: Evidence from U.S. Thrift Institutions <i>Robert DeYoung and Evren Örs</i>	WP-04-25
Monetary Policy with State Contingent Interest Rates Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-26
Comparing location decisions of domestic and foreign auto supplier plants Thomas Klier, Paul Ma and Daniel P. McMillen	WP-04-27
China's export growth and US trade policy Chad P. Bown and Meredith A. Crowley	WP-04-28
Where do manufacturing firms locate their Headquarters? J. Vernon Henderson and Yukako Ono	WP-04-29
Monetary Policy with Single Instrument Feedback Rules Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-30

Firm-Specific Capital, Nominal Rigidities and the Business Cycle David Altig, Lawrence J. Christiano, Martin Eichenbaum and Jesper Linde	WP-05-01
Do Returns to Schooling Differ by Race and Ethnicity? Lisa Barrow and Cecilia Elena Rouse	WP-05-02
Derivatives and Systemic Risk: Netting, Collateral, and Closeout Robert R. Bliss and George G. Kaufman	WP-05-03
Risk Overhang and Loan Portfolio Decisions Robert DeYoung, Anne Gron and Andrew Winton	WP-05-04
Characterizations in a random record model with a non-identically distributed initial record <i>Gadi Barlevy and H. N. Nagaraja</i>	WP-05-05
Price discovery in a market under stress: the U.S. Treasury market in fall 1998 Craig H. Furfine and Eli M. Remolona	WP-05-06
Politics and Efficiency of Separating Capital and Ordinary Government Budgets Marco Bassetto with Thomas J. Sargent	WP-05-07
Rigid Prices: Evidence from U.S. Scanner Data Jeffrey R. Campbell and Benjamin Eden	WP-05-08
Entrepreneurship, Frictions, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-05-09
Wealth inequality: data and models Marco Cagetti and Mariacristina De Nardi	WP-05-10
What Determines Bilateral Trade Flows? Marianne Baxter and Michael A. Kouparitsas	WP-05-11
Intergenerational Economic Mobility in the U.S., 1940 to 2000 Daniel Aaronson and Bhashkar Mazumder	WP-05-12
Differential Mortality, Uncertain Medical Expenses, and the Saving of Elderly Singles Mariacristina De Nardi, Eric French, and John Bailey Jones	WP-05-13
Fixed Term Employment Contracts in an Equilibrium Search Model Fernando Alvarez and Marcelo Veracierto	WP-05-14
Causality, Causality, Causality: The View of Education Inputs and Outputs from Economics Lisa Barrow and Cecilia Elena Rouse	WP-05-15

Competition in Large Markets Jeffrey R. Campbell	WP-05-16
Why Do Firms Go Public? Evidence from the Banking Industry Richard J. Rosen, Scott B. Smart and Chad J. Zutter	WP-05-17
Clustering of Auto Supplier Plants in the U.S.: GMM Spatial Logit for Large Samples <i>Thomas Klier and Daniel P. McMillen</i>	WP-05-18
Why are Immigrants' Incarceration Rates So Low? Evidence on Selective Immigration, Deterrence, and Deportation <i>Kristin F. Butcher and Anne Morrison Piehl</i>	WP-05-19
Constructing the Chicago Fed Income Based Economic Index – Consumer Price Index: Inflation Experiences by Demographic Group: 1983-2005 Leslie McGranahan and Anna Paulson	WP-05-20
Universal Access, Cost Recovery, and Payment Services Sujit Chakravorti, Jeffery W. Gunther, and Robert R. Moore	WP-05-21
Supplier Switching and Outsourcing Yukako Ono and Victor Stango	WP-05-22
Do Enclaves Matter in Immigrants' Self-Employment Decision? Maude Toussaint-Comeau	WP-05-23
The Changing Pattern of Wage Growth for Low Skilled Workers Eric French, Bhashkar Mazumder and Christopher Taber	WP-05-24
U.S. Corporate and Bank Insolvency Regimes: An Economic Comparison and Evaluation Robert R. Bliss and George G. Kaufman	WP-06-01
Redistribution, Taxes, and the Median Voter Marco Bassetto and Jess Benhabib	WP-06-02
Identification of Search Models with Initial Condition Problems Gadi Barlevy and H. N. Nagaraja	WP-06-03
Tax Riots Marco Bassetto and Christopher Phelan	WP-06-04
The Tradeoff between Mortgage Prepayments and Tax-Deferred Retirement Savings Gene Amromin, Jennifer Huang, and Clemens Sialm	WP-06-05
Why are safeguards needed in a trade agreement? Meredith A. Crowley	WP-06-06

Taxation, Entrepreneurship, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-06-07
A New Social Compact: How University Engagement Can Fuel Innovation Laura Melle, Larry Isaak, and Richard Mattoon	WP-06-08
Mergers and Risk Craig H. Furfine and Richard J. Rosen	WP-06-09
Two Flaws in Business Cycle Accounting Lawrence J. Christiano and Joshua M. Davis	WP-06-10
Do Consumers Choose the Right Credit Contracts? Sumit Agarwal, Souphala Chomsisengphet, Chunlin Liu, and Nicholas S. Souleles	WP-06-11
Chronicles of a Deflation Unforetold François R. Velde	WP-06-12
Female Offenders Use of Social Welfare Programs Before and After Jail and Prison: Does Prison Cause Welfare Dependency? <i>Kristin F. Butcher and Robert J. LaLonde</i>	WP-06-13
Eat or Be Eaten: A Theory of Mergers and Firm Size Gary Gorton, Matthias Kahl, and Richard Rosen	WP-06-14
Can Bonds Hedge Volatility Risk in the U.S. Treasury Market? A Specification Test for Affine Term Structure Models <i>Torben G. Andersen and Luca Benzoni</i>	WP-06-15
Transforming Payment Choices by Doubling Fees on the Illinois Tollway Gene Amromin, Carrie Jankowski, and Richard D. Porter	WP-06-16
How Did the 2003 Dividend Tax Cut Affect Stock Prices? Gene Amromin, Paul Harrison, and Steven Sharpe	WP-06-17
Will Writing and Bequest Motives: Early 20th Century Irish Evidence Leslie McGranahan	WP-06-18
How Professional Forecasters View Shocks to GDP Spencer D. Krane	WP-06-19
Evolving Agglomeration in the U.S. auto supplier industry Thomas Klier and Daniel P. McMillen	WP-06-20
Mortality, Mass-Layoffs, and Career Outcomes: An Analysis using Administrative Data Daniel Sullivan and Till von Wachter	WP-06-21

The Agreement on Subsidies and Countervailing Measures: Tying One's Hand through the WTO. <i>Meredith A. Crowley</i>	WP-06-22
How Did Schooling Laws Improve Long-Term Health and Lower Mortality? Douglas Almond and Bhashkar Mazumder	WP-06-23
Manufacturing Plants' Use of Temporary Workers: An Analysis Using Census Micro Data Yukako Ono and Daniel Sullivan	WP-06-24
What Can We Learn about Financial Access from U.S. Immigrants? Una Okonkwo Osili and Anna Paulson	WP-06-25
Bank Imputed Interest Rates: Unbiased Estimates of Offered Rates? Evren Ors and Tara Rice	WP-06-26